\documentclass{ws-rv9x6}
\usepackage{ws-rv-van}             % numbered citation/references (default)
%\usepackage{ws-index}             % to produce multiple indices
\makeindex
%\newindex{aindx}{adx}{and}{Author Index}       % author index
%\renewindex{default}{idx}{ind}{Subject Index}  % subject index
\begin{document}
\begin{thebibliography}{25}
\providecommand{\natexlab}[1]{#1}
\providecommand{\url}[1]{\texttt{#1}}
\expandafter\ifx\csname urlstyle\endcsname\relax
  \providecommand{\doi}[1]{doi: #1}\else
  \providecommand{\doi}{doi: \begingroup \urlstyle{rm}\Url}\fi

\bibitem{laplace-prob}
P.-S. Laplace, \emph{Philosophical Essay on Probabilities. {T}ranslated from
  the fifth {F}rench edition of 1825}. (Springer, Berlin, New York, 1995,1998).
\newblock ISBN 978-0-387-94349-7.
\newblock URL \url{http://www.archive.org/details/philosophicaless00lapliala}.

\bibitem{poincare14}
H.~Poincar{\'{e}}, \emph{Wissenschaft und Hypothese}. (Teubner, Leipzig, 1914).

\bibitem{Sundman12}
K.~E. Sundman, Memoire sur le probl{\`{e}}me de trois corps, \emph{Acta
  Mathematica}. {\bf 36}, \penalty0 105--179,  (1912).

\bibitem{Wang91}
Q.~D. Wang, The global solution of the $n$-body problem, \emph{Celestial
  Mechanics}. {\bf 50}, \penalty0 73--88,  (1991).
\newblock \doi{10.1007/BF00048987}.
\newblock URL \url{http://dx.doi.org/10.1007/BF00048987}.

\bibitem{Diacu96}
F.~Diacu, The solution of the n-body problem, \emph{The Mathematical
  Intelligencer}. {\bf 18}\penalty0 (3), \penalty0 66--70,  (1996).

\bibitem{Wang01}
Q.~D. Wang, Power series solutions and integral manifold of the n-body problem,
  \emph{Regular \& Chaotic Dynamics}. {\bf 6}\penalty0 (4), \penalty0 433--442,
   (2001).
\newblock \doi{10.1070/RD2001v006n04ABEH000187}.
\newblock URL \url{http://dx.doi.org/10.1070/RD2001v006n04ABEH000187}.

\bibitem{fred-tof-82}
E.~Fredkin and T.~Toffoli, Conservative logic, \emph{International Journal of
  Theoretical Physics}. {\bf 21}\penalty0 (3-4), \penalty0 219--253,  (1982).
\newblock \doi{10.1007/BF01857727}.
\newblock URL \url{http://dx.doi.org/10.1007/BF01857727}.
\newblock reprinted in \cite[Part I, Chapter 3]{adama02}.

\bibitem{margolus-billard}
N.~Margolus, Physics-like model of computation, \emph{Physica}. {\bf D10},
  \penalty0 81--95,  (1984).
\newblock reprinted in \cite[Part I, Chapter 4]{adama02}.

\bibitem{margolus-02}
N.~Margolus.
\newblock Universal cellular automata based on the collisions of soft spheres.
\newblock In ed. A.~Adamatzky, \emph{Collision-based computing}, pp. 107--134.
  Springer, London,  (2002).
\newblock URL \url{http://people.csail.mit.edu/nhm/cca.pdf}.

\bibitem{Adamatzky-02}
A.~Adamatzky.
\newblock New media for collision-based computing.
\newblock In ed. A.~Adamatzky, \emph{Collision-based computing}, pp. 411--442.
  Springer, London,  (2002).
\newblock URL \url{http://people.csail.mit.edu/nhm/cca.pdf}.

\bibitem{go-67}
E.~M. Gold, Language identification in the limit, \emph{Information and
  Control}. {\bf 10}, \penalty0 447--474,  (1967).
\newblock \doi{10.1016/S0019-9958(67)91165-5}.
\newblock URL \url{http://dx.doi.org/10.1016/S0019-9958(67)91165-5}.

\bibitem{blum75blum}
L.~Blum and M.~Blum, Toward a mathematical theory of inductive inference,
  \emph{Information and Control}. {\bf 28}\penalty0 (2), \penalty0 125--155
  (June, 1975).

\bibitem{angluin:83}
D.~Angluin and C.~H. Smith, A survey of inductive inference: Theory and
  methods, \emph{Computing Surveys}. {\bf 15}, \penalty0 237--269,  (1983).

\bibitem{ad-91}
L.~M. Adleman and M.~Blum, Inductive inference and unsolvability, \emph{The
  Journal of Symbolic Logic}. {\bf 56}, \penalty0 891--900 (Sept., 1991).
\newblock \doi{10.2307/2275058}.
\newblock URL \url{http://dx.doi.org/10.2307/2275058}.

\bibitem{li:92}
M.~Li and P.~M.~B. Vit{\'{a}}nyi, Inductive reasoning and {K}olmogorov
  complexity, \emph{Journal of Computer and System Science}. {\bf 44},
  \penalty0 343--384,  (1992).
\newblock \doi{10.1016/0022-0000(92)90026-F}.
\newblock URL \url{http://dx.doi.org/10.1016/0022-0000(92)90026-F}.

\bibitem{rogers1}
H.~{Rogers, Jr.}, \emph{Theory of Recursive Functions and Effective
  Computability}. (MacGraw-Hill, New York, 1967).

\bibitem{odi:89}
P.~Odifreddi, \emph{Classical Recursion Theory, Vol. 1}. (North-Holland,
  Amsterdam, 1989).

\bibitem{odi:99}
P.~Odifreddi, \emph{Classical Recursion Theory, Vol. 2}. (North-Holland,
  Amsterdam, 1999).

\bibitem{chaitin3}
G.~J. Chaitin, \emph{Algorithmic Information Theory}. (Cambridge University
  Press, Cambridge, 1987).

\bibitem{calude:94}
C.~Calude, \emph{Information and Randomness---An Algorithmic Perspective}.
  (Springer, Berlin, 1994).

\bibitem{calude-dinneen06}
C.~S. Calude and M.~J. Dinneen.
\newblock Exact approximations of omega numbers.
\newblock URL
  \url{http://www.cs.auckland.ac.nz/CDMTCS//researchreports/293crismjd.pdf}.
\newblock CDMTCS report series 293,  (2006).

\bibitem{Specker49}
E.~Specker, Nicht konstruktiv beweisbare {S}\"atze der {A}nalysis, \emph{The
  Journal of Smbolic Logic}. {\bf 14}, \penalty0 145--158,  (1949).
\newblock Reprinted in \cite[pp. 35--48]{specker-ges}; {E}nglish translation:
  {\it Theorems of Analysis which cannot be proven constructively}.

\bibitem{Specker57}
E.~Specker.
\newblock Der {S}atz vom {M}aximum in der rekursiven {A}nalysis.
\newblock In ed. A.~Heyting, \emph{Constructivity in mathematics : proceedings
  of the colloquium held at Amsterdam, 1957}, pp. 254--265, Amsterdam,  (1959).
  North-Holland Publishing Company.
\newblock Reprinted in \cite[pp. 148--159]{specker-ges}; {E}nglish translation:
  {\it Theorems of Analysis which cannot be proven constructively}.

\bibitem{adama02}
A.~Adamatzky, \emph{Collision-based computing}. (Springer, London, 2002).

\bibitem{specker-ges}
E.~Specker, \emph{Selecta}. (Birkh{\"{a}}user Verlag, Basel, 1990).

\end{thebibliography}


\chapter{Omega and the time evolution of the $n$-body problem\label{chapter-svozil}}
%\footnote{Contribution to the collection of papers ``Randomness \& Complexity, from Leibniz to Chaitin,'' ed. by Cristian S. Calude}}

\author[K. Svozil]{Karl Svozil}
\address{Institut f\"ur Theoretische Physik, University of Technology Vienna,\\
Wiedner Hauptstra\ss e 8-10/136, A-1040 Vienna, Austria\\
svozil@tuwien.ac.at}


\begin{abstract}
The series solution of the behavior of a finite number of physical bodies and Chaitin's Omega number
share quasi-algorithmic expressions; yet both lack a computable radius of convergence.
\end{abstract}


%\pacs{01.70.+w,01.65.+g,02.30.Lt,03.65.Ta}
%\keywords{Unknowables, Church-Turing thesis, induction and forecast, n-body problem, halting probability, Omega number}

\body


\section{Solutions to the $n$--body problem}

The behaviour and evolution of a finite number of bodies is a sort of ``rosetta stone''
of classical celestial mechanics insofar as its investigation induced a lot of
twists, revelations and unexpected issues.
Arguably the most radical deterministic position on the subject
was formulated by Laplace,
stating that \cite[Chapter II]{laplace-prob}
%\begin{quote}
{\em
``We ought then to regard the present state of the
universe as the effect of its anterior state and as the
cause of the one which is to follow. Given for one
instant an intelligence which could comprehend all the
forces by which nature is animated and the respective
situation of the beings who compose it an intelligence
sufficiently vast to submit these data to analysis it
would embrace in the same formula the movements of
the greatest bodies of the universe and those of the
lightest atom; for it, nothing would be uncertain and
the future, as the past, would be present to its eyes.''
}
%\end{quote}

In what may be considered as the beginning of deterministic chaos theory,
Poincar{\'e} was forced to accept a gradual departure from the deterministic position:
sometimes small
variations in the initial state of the bodies could lead to huge variations in their
evolution in later times.
In Poincar{\'e}'s own words  \cite[Chapter 4, Sect. II, pp. 56-57]{poincare14},
{\em
``If we would know the laws of Nature and the state of the Universe precisely
for a certain time,
we would be able to predict with certainty
the state of the Universe for any later time.
But
[[~$\ldots$~]]
it can be the case that small differences in the initial values
produce great differences in the later phenomena;
a small error in the former may result in a large error in the latter.
The prediction becomes impossible and we have a `random phenomenon.'
''}


In what follows we present an even more radical departure from Laplacian determinism.
A physical system of a finite number of bodies capable of universal computation will be presented
which has the property that certain propositions remain not only provable
intractable, but provable unknowable.
Pointedly stated, our knowledge of any such system remains incomplete forever.
For the sake of making things worse, we shall ``compress'' and ``compactify''
this kind of physical incompleteness by considering physical observables
which are truly random, i.e., algorithmically incompressible and stochastic.

The methods of construction of
physical $n$--body observables exhibiting the above features turn out to be rather humble and straightforward.
In a first step, it suffices to reduce the problem to the halting problem for universal computation.
This can be achieved by ``embedding'' a universal computer into a
suitable physical system of a finite number of bodies.
The associated ballistic computation will be presented in the next section.
In a second reduction step, the universal computer will be directed to attempt to ``compute''
Chaitin's Omega number, which is provable random,
and which is among the ``most difficult'' tasks imaginable.
Finally, consequences for the series solutions
 \cite{Sundman12,Wang91,Diacu96,Wang01} to the general $n$-body problem will be discussed.



\section{Reduction by ballistic computation}

In order to embed reversible universal computation into a quasi-physical environment,
Fredkin and Toffoli introduced a ``billiard ball model''
\cite{fred-tof-82,margolus-billard,margolus-02,Adamatzky-02}
based on the collisions of spheres as well as on mirrors reflecting the spheres.
Thus collisions and reflections are the basic ingredients for building universal computation.

If we restrict ourselves to classical gravitational potentials without collisions,
we do not have any repulsive
interaction at our disposal; only attractive $1/r$ potentials.
Thus the kinematics corresponding to reflections and collisions has to be realized
by purely attractive interactions.
Fig.~\ref{2007-chaitin-f1}a) depicts a Fredkin gate realized by attractive interaction
which corresponds to the analogue billiard ball configuration achieved by collisions
(e.g., \cite[Fig.~4.5]{margolus-billard}).
At points
${\bf A}$
and
${\bf B}$
and time $t_i$,
two bodies are either put at both locations ${\bf A}$
and
${\bf B}$; or alternatively, one body is put
at only one location, or no bodies are placed at all.
If bodies are present at both  ${\bf A}$
and
${\bf B}$, then they will follow the right paths at later times $t_f$.
In case only one body is present at ${\bf A}$
or
${\bf B}$, only one of the dotted inner outgoing paths will be used.
Boolean logic can be implemented by the presence or absence of balls.
Fig.~\ref{2007-chaitin-f1}b) depicts a reflective ``mirror'' element realized by a quasi-steady mass.
\begin{figure*}
\begin{center}
\begin{tabular}{ccccccc}
%TeXCAD Picture [1.pic]. Options:
%\grade{\on}
%\emlines{\off}
%\epic{\off}
%\beziermacro{\on}
%\reduce{\on}
%\snapping{\off}
%\quality{8.00}
%\graddiff{0.01}
%\snapasp{1}
%\zoom{4.0000}
\unitlength .4mm % = 1.14pt
\linethickness{0.4pt}
\ifx\plotpoint\undefined\newsavebox{\plotpoint}\fi % GNUPLOT compatibility
\begin{picture}(115,130)(0,0)
\qbezier(10,10)(30.5,80.5)(60,80)
\qbezier(10,110)(30.5,39.5)(60,40)
\qbezier(109.5,10)(89,80.5)(59.5,80)
\qbezier(109.5,110)(89,39.5)(59.5,40)
%\vector{dash}{1}(34.25,94.5)(35.5,98.25)
\put(35.5,98.25){\vector(1,3){.18}}\multiput(34.07,94.32)(.0833,.25){3}{\line(0,1){.25}}
\multiput(34.57,95.82)(.0833,.25){3}{\line(0,1){.25}}
\multiput(35.07,97.32)(.0833,.25){3}{\line(0,1){.25}}
%\end
%\vector{dash}{1}(11.25,13.75)(12.25,17.5)
\put(12.25,17.5){\vector(1,4){.18}}\multiput(11.07,13.57)(.0667,.25){3}{\line(0,1){.25}}
\multiput(11.47,15.07)(.0667,.25){3}{\line(0,1){.25}}
\multiput(11.87,16.57)(.0667,.25){3}{\line(0,1){.25}}
%\end
%\vector{dash}{1}(11.25,106.25)(12.25,102.5)
\put(12.25,102.5){\vector(1,-4){.18}}\multiput(11.07,106.07)(.0667,-.25){3}{\line(0,-1){.25}}
\multiput(11.47,104.57)(.0667,-.25){3}{\line(0,-1){.25}}
\multiput(11.87,103.07)(.0667,-.25){3}{\line(0,-1){.25}}
%\end

%\vector{dash}{1}(34,27)(34.25,25.25)
\put(34.25,25.25){\vector(1,-4){.18}}\multiput(33.82,26.82)(.0625,-.4375){2}{\line(0,-1){.4375}}
%\end
%\dashline{1}(10,10)(38.5,110)
\multiput(9.82,9.82)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(10.37,11.73)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(10.91,13.63)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(11.45,15.54)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(12,17.44)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(12.54,19.35)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(13.08,21.25)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(13.62,23.16)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(14.17,25.06)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(14.71,26.97)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(15.25,28.87)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(15.8,30.78)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(16.34,32.68)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(16.88,34.59)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(17.42,36.49)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(17.97,38.4)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(18.51,40.3)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(19.05,42.21)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(19.6,44.11)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(20.14,46.01)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(20.68,47.92)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(21.22,49.82)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(21.77,51.73)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(22.31,53.63)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(22.85,55.54)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(23.4,57.44)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(23.94,59.35)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(24.48,61.25)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(25.02,63.16)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(25.57,65.06)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(26.11,66.97)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(26.65,68.87)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(27.2,70.78)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(27.74,72.68)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(28.28,74.59)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(28.82,76.49)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(29.37,78.4)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(29.91,80.3)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(30.45,82.21)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(31,84.11)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(31.54,86.01)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(32.08,87.92)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(32.62,89.82)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(33.17,91.73)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(33.71,93.63)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(34.25,95.54)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(34.8,97.44)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(35.34,99.35)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(35.88,101.25)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(36.42,103.16)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(36.97,105.06)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(37.51,106.97)(.0679,.2381){4}{\line(0,1){.2381}}
\multiput(38.05,108.87)(.0679,.2381){4}{\line(0,1){.2381}}
%\end
%\dashline{1}(10,110)(38.5,10)
\multiput(9.82,109.82)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(10.37,107.92)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(10.91,106.01)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(11.45,104.11)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(12,102.21)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(12.54,100.3)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(13.08,98.4)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(13.62,96.49)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(14.17,94.59)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(14.71,92.68)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(15.25,90.78)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(15.8,88.87)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(16.34,86.97)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(16.88,85.06)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(17.42,83.16)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(17.97,81.25)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(18.51,79.35)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(19.05,77.44)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(19.6,75.54)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(20.14,73.63)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(20.68,71.73)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(21.22,69.82)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(21.77,67.92)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(22.31,66.01)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(22.85,64.11)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(23.4,62.21)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(23.94,60.3)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(24.48,58.4)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(25.02,56.49)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(25.57,54.59)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(26.11,52.68)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(26.65,50.78)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(27.2,48.87)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(27.74,46.97)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(28.28,45.06)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(28.82,43.16)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(29.37,41.25)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(29.91,39.35)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(30.45,37.44)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(31,35.54)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(31.54,33.63)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(32.08,31.73)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(32.62,29.82)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(33.17,27.92)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(33.71,26.01)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(34.25,24.11)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(34.8,22.21)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(35.34,20.3)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(35.88,18.4)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(36.42,16.49)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(36.97,14.59)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(37.51,12.68)(.0679,-.2381){4}{\line(0,-1){.2381}}
\multiput(38.05,10.78)(.0679,-.2381){4}{\line(0,-1){.2381}}
%\end
\put(0,55){\makebox(0,0)[cc]{$t_i$}}
\put(120,55){\makebox(0,0)[cc]{$t_f$}}
%
\put(0,109){\makebox(0,0)[cc]{${\bf A}$?}}
\put(0,10){\makebox(0,0)[cc]{${\bf B}$?}}
\put(120,110){\makebox(0,0)[cc]{${\bf AB}$}}
\put(120,10){\makebox(0,0)[cc]{${\bf AB}$}}
\put(50,110){\makebox(0,0)[cc]{$\overline{{\bf A}}{\bf B}$}}
\put(50,10){\makebox(0,0)[cc]{${\bf A}\overline{{\bf B}}$}}
%\vector(106.75,101.5)(107.25,103.25)
\put(107.25,103.25){\vector(1,4){.18}}\multiput(106.75,101.5)(.08333,.29167){6}{\line(0,1){.29167}}
%\end
%\vector(105.75,22)(106.25,20)
\put(106.25,20){\vector(1,-4){.18}}\multiput(105.75,22)(.08333,-.33333){6}{\line(0,-1){.33333}}
%\end
\end{picture}
&$\qquad $&
%TeXCAD Picture [2.pic]. Options:
%\grade{\on}
%\emlines{\off}
%\epic{\off}
%\beziermacro{\on}
%\reduce{\on}
%\snapping{\off}
%\quality{8.00}
%\graddiff{0.01}
%\snapasp{1}
%\zoom{4.0000}
\unitlength .4mm % = 1.14pt
\linethickness{0.4pt}
\ifx\plotpoint\undefined\newsavebox{\plotpoint}\fi % GNUPLOT compatibility
\begin{picture}(115,80.5)(0,0)
\qbezier(10,10)(30.5,80.5)(60,80)
\qbezier(109.5,10)(89,80.5)(59.5,80)
%\vector{dash}{1}(11.25,13.75)(12.25,17.5)
\put(12.25,17.5){\vector(1,4){.18}}\multiput(11.07,13.57)(.0667,.25){3}{\line(0,1){.25}}
\multiput(11.47,15.07)(.0667,.25){3}{\line(0,1){.25}}
\multiput(11.87,16.57)(.0667,.25){3}{\line(0,1){.25}}
%\end

\put(0,10){\makebox(0,0)[cc]{${\bf A}$}}
\put(120,10){\makebox(0,0)[cc]{${\bf A}$}}
%\vector(105.75,22)(106.25,20)
\put(106.25,20){\vector(1,-4){.18}}\multiput(105.75,22)(.08333,-.33333){6}{\line(0,-1){.33333}}
%\end
\put(60,40){\circle*{10}}
\end{picture}
\\
a)&&b)
\end{tabular}
\end{center}
\caption{Elements of universal ballistic computation realized by
attractive $1/r$ potentials.
a) Fredkin's gate can perform logical reversibility:
bodies will appear on the right outgoing paths
if and only if bodies came in at both ${\bf A}$ and ${\bf B}$;
b) Reflective ``mirror'' element realized by a quasi-steady mass.
\label{2007-chaitin-f1}}
\end{figure*}
For a proof of universality, we refer to the classical papers
on the billiard ball model cited above.



\section{Undecidability and Omega in the $n$-body problem}

By reduction to the recursive unsolvability
of the rule inference \cite{go-67,blum75blum,angluin:83,ad-91,li:92}
and the halting \cite{rogers1,odi:89,odi:99} problems,
the general induction and forecasting problem of the
$n$-body ballistic universal computer sketched above is provable unsolvable.
That is, there exist initial configurations for which it
is impossible to predict with certainty whether or not certain
``final'' states will eventually be reached.
Moreover, given a finite segment of the time evolution alone
is in general insufficient for a derivation of the initial state configuration
of the $n$-body problem.


For the sake of making things worse, we imagine an $n$-body system
attempting to evaluate its associated halting probability Omega
\cite{chaitin3,calude:94,calude-dinneen06}.
In order to establish the equivalent of prefix-free programs,
only a limited number of $n$-body initial configurations contribute to
the configuration.
Furthermore, as the computation is reversible and procedural,
certain ``final'' configurations must be defined as halting states.
This is a feature shared with the billiard ball model,
as well as with quantum computation.


\section{Consequences for series solutions}

Wang's power series solution to the $n$-body problem \cite{Wang91,Wang01}
may converge ``very slowly'' \cite{Diacu96}.
Indeed, by considering the halting problems above, and
in particular by reduction to the computation of the halting probability Omega,
certain physical observables associated with the $n$-body problem
do not have a power series solution with a {\em computable radius of convergence}.

This is a particular case of Specker's theorems in recursive analysis, stating that
there exist recursive monotone bounded sequences of rational numbers
whose limit is no computable number
\cite{Specker49}; and
there exist a recursive real function which has its maximum in the unit interval
at no recursive real number \cite{Specker57}.

It is important to realize that,
while it may be possible to evaluate
the state of the $n$ bodies by Wang's power series solution
for any finite time with a computable,
though excessively large, radius of convergence,
global observables, referring to all times, may be uncomputable.
Examples of global observables are, for instance, associated
with the stability of the solar system and associated with it,
bounds for the orbits.

This, of course,
stems from the metaphor and robustness of universal computation
and the capacity of the $n$-body problem to implement universality.
It is no particularity and peculiarity of Wang's power series solution.
Indeed, the troubles reside in the capabilities to implement Peano arithmetic and
universal computation by $n$-body problems.
Because of this capacity, there cannot exist other formalizable methods,
analytic solutions or approximations capable to decide and compute certain decision problems
or observables for the $n$-body problem.

Chaitin's Omega number,
the halting probability for universal computers,
has been invented in a totally different, unrelated algorithmic context,
and with intentions in mind which are seemingly different from issues in classical mechanics.
Thus it is fascinating that Omega is also relevant for the prediction of the behaviour
and the movement of celestial bodies.



%\bibliographystyle{ws-rv-van}
%\bibliography{svozil}




\end{document}

